Author's response to reviews

Title: Altered motor control patterns in whiplash and chronic neck pain.

Authors:

Astrid Woodhouse (astrid.woodhouse@ntnu.no)
Ottar Vasseljen (ottar.vasseljen@ntnu.no)

Version: 4 Date: 17 June 2008

Author's response to reviews: see over
Dear Editor,

Thank you for having accepted our manuscript “Altered motor control patterns in whiplash and chronic neck pain”. Some formatting changes were requested and have now been addressed in a revised version:

1. The authors in the manuscript are now written exactly as in the submission system (forename, surname) and also in the order in which they are listed in the submission system.

2. The authors’ qualifications have been removed from the manuscript

3. The reference list now contains the names of all authors in cases where the term “et al” was previously used (reference numbers 16, 24 and 35)

4. All tables are now fitted to portrait A4 page format.

5. Both figures have been cropped to minimize white space around the image.

Sincerely yours,

__________________________  ______________________
Astrid Woodhouse            Ottar Vasseljen